Author's response to reviews

Title: Splash!: an innovative prospective cohort study of the impact of environmental, social and family-level influences on child oral health and obesity related risk factors and outcomes.

Authors:

Andrea M de Silva-Sanigorski (andreams@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)
Hanny Calache (calacheh@dhs.v.org.au)
Michael Smith (MICHAEL@BarwonHealth.org.au)
Lisa Gold (lisa.gold@deakin.edu.au)
Mark Gussy (M.Gussy@latrobe.edu.au)
Anthony Scott (a.scott@unimelb.edu.au)
Monica Virgo-Milton (monicav@unimelb.edu.au)

Version: 2 Date: 19 June 2011

Author's response to reviews:

Response to Reviewers comments

The revised manuscript now contains the additional information as requested, as detailed below.

(1) Document, within your manuscript, the name of the committee which granted ethics approval for your study.

This information has now been added (line 442-444, page 20)

(2) Provide, within your manuscript, a power calculation used to determine the sample size of your study.

This information has now been added (line 197-203, page 10)

(3) Document, within your manuscript, whether you will be receiving informed consent from your study participants and whether this will be verbal or written.

This information has now been added (line 444-446, page 20)